VC/ 9-Mar-04
Week 11  starting 08 March 04

SM18 Operation Meeting (Tuesdays)

 Bench status  and tests for this week : 

	Bench
	Name
	
	Temp
	-
	-

	TBC1
	MBBR2052
	NEW 
	 
	 
	 

	TBC2
	SSS002
	EX1 Extra Test
	 
	 
	 

	TBD1
	MBBL3071
	PT 1 IAP @ cold
	 
	 
	 

	TBD2
	MBBR3063
	NEW
	 
	 
	 

	TBE1
	MBAL3105
	ICS 3 
	 
	 
	 

	TBE2
	MBAL1056
	NEW
	 
	 
	 

	TBF1
	MBBR1041
	PT 12 Warm Up
	 
	 
	 

	TBF2
	MBBR2014
	CR 1 Leak Test
	 
	 
	 


Test Coordinator this week:  Pierre.Pugnat  ;  MTM  Tech. on call :  (16-3602) . 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. Priority List in SMCR includes shafts/no shafts

2. AOB

	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday_02_A

Saturday_06_M


	TBE1:C11 pins damaged on magnet 1054. Bad manipulation by ICS during HV test 

Same on 2027 (TBD2)
	ICS
	
	Richard replaced the connectors. OK But no spare now. ICS responsible aware.

	Tuesday_02_A


	TBE1:HF not getting armed.
	MTM
	HF, LF
	HF, LF switched off. Crates powered off. OK

	Tuesday_02_A


	Mobic was not functioning correctly. Windows was hung.
	MTM
	
	Richard advised to follow the procedure written by Maryline :Reset the program. Then OK

	Wednesday_03_M


	TBF2 : IAP @cold . HRWREC mentioned that a fixed equipment was not recognized (the generator for AC compensation)
	MTM
	
	Solved by Federico.

Keep the equipment in place.

	Wednesday_03_A


	TBD1: Trigger not received by CTH during a quench (mentioned last week)
	ACR
	
	Process corrected by A.Tovar.

	Thursday_04_N


	TBF1: IAP @warm : Permanent trigger originating from the ext-int V clamp card.
	ICS
	
	Richard contacted. He called ICS team leader. It was found that the soldering of the Vtaps VT1_5, VT2_5 in the bus bar were not correct. Then OK

	Thursday_04_N


	TBC2; Cryo problem : quench not possible
	ACR
	
	Problem with cold compressor.

	Friday_05_A


	TBF2: IAP final at warm. LF, HF DAQ not get armed. Reset all the crates. Exit from TEMA. When doing exit: meassage  : wait a moment. TEMA blocked.
	MTM/AB-CO
	TEMA
	Richard contacted. He called Magnus who killed the process.

A.Raimondo is preparing a tool to be used by equipment support to do it.

	Friday_05_A


	TBE1: End of HV test, quench signal received because of A.Tovar accidentally stopped the Communication PLC/CTH. CFB pressure increased ????.
	ACR
	
	No explanation for this increase of pressure.

Valve was closed ΔT ->11k

	Friday_05_N


	TBC1: Unable to adjust amplitude of D2_U-L below  77 mV (AC comp cold)

TBD1: Same for D1 l-U below 100 mV

Initially D20 was not connected and 9v at potaim card
	MTM
	IAP
	Advise of Richard. Check if all the connectors are correctly plugged in particular the D20.

Revise IAP criteria.

	Friday_05_N


	TBE1:Min energy launched from TEMA . Ramped to 11850 A the ramp down after 20 s then ramp up without intervention. 
	MTM/ABCO
	TEMA
	Cycle is OK (checked by Maryline).

Slow PA

	Friday_05_N


	TBC1: no current reading in Aperture 1 

One cable remains plugged in the SSS crates.
	MTM
	
	Richard solved the problem.

	Friday_05_N


	TBC1: Quench at 1.5 kA Error in AQA.

Triggered by Vbus.
	MTM
	
	Symmetric quench?

	Saturday_06_N


	TBC1: During LHC cycle, VME error. Cycle interrupted 

The auto test was good.
	AB/CO
	
	Richard advises to reset VME crates then repeat a small quench and re-start LHC cycle. He asked to confirm with the test coordinator. Test coordinator decided to stop the magnetic measurements because VME errors happened frequently on C1 and magnet was good.

Should we change the crates?

	Saturday_06_N


	TBC1: HV test at cold stopped four times

Message read : The megger stopped the measurement.

Internal problem of Megger
	MTM
	
	Perform run-recovery.

Next version of HV : Automatic run recovery implemented in that case. After filtering errors.

	Sunday_07_N


	TBF2 (2014 warm): In the traveller is is mentioned that the times when the electrical test done by MTM are not reported then delaying the test sequences. This is not the case. The traveller was filled in time (Kevin)
	MTM/ACR/ICS.
	
	








